

Author's response to reviews

Title: The Impact of Bone Morphology on the Outcome of the Pivot Shift Test: A Cohort Study

Authors:

Thomas Branch (doctorbranch@yahoo.com)

Shaun Stinton (stinton@gmail.com)

Adrija Sharma (adrija.sharma@gmail.com)

Frederic Lavoie (fredericlavoie.md@gmail.com)

Christian Guier (guier@jhmail.com)

Philippe Neyret (philippe.neyret@chu-lyon.fr)

Version: 4 **Date:** 31 Aug 2017

Author's response to reviews:

Unfortunately your current Ethics statement is unacceptable.

Please revise your Ethics section to include the FULL, OFFICIAL NAME of the ethics committee that approved your study, and ethical approval number, if applicable.

In response to the comment above, we have stated in the methods section of the paper that no IRB approval was required at the named institute. We cannot provide an official name of an ethics committee because no such approval was required when the data was collected in 2008. Informed consent was obtained as described in the Methods.